DOCTORAL SCHOOL IN NATURAL SCIENCES DISSERTATION SERIES M 4/2022

PARAMETERISATION OF SEA ICE COVER In Short-Range Regional Numerical Weather Prediction

Yurii Batrak

Faculty of Science Doctoral Programme in Atmospheric Sciences University of Helsinki Helsinki, Finland

Academic dissertation

To be presented, with the permission of the Faculty of Science of the University of Helsinki, for public criticism in the auditorium E204 of Physicum, Gusłaf Hällströmin katu 2, on August 24;th 2022, at 1 o'clock in the afternoon.

Helsinki 2022

Author's Address	Development Centre for Weather Forecasting Norwegian Meteorological Institute P.O. Box 43 Blindern NO-0313 Oslo, Norway email yurii.batrak@met.no						
Supervisor	Professor Heikki Järvinen Institute for Atmospheric and Earth System Research University of Helsinki, Finland						
Pre-examiners	Research Professor David Bromwich Atmospheric Sciences Program, Department of Geography The Ohio State University, USA						
	Research Professor Jari Haapala Marine Research Unit Finnish Meteorological Institute, Finland						
Opponent	Senior Scientist Steffen Tietsche Research Department, Earth System Predictability Section European Centre for Medium-Range Weather Forecasts Bonn, Germany						
Custos	Professor Heikki Järvinen Institute for Atmospheric and Earth System Research University of Helsinki, Finland						

ISBN 978-951-51-8373-6 (softcover) ISSN 2669-882X Unigrafia Oy Helsinki 2022

ISBN 978-951-51-8374-3 (PDF) ISSN 2670-2010 Helsinki 2022 http://ethesis.helsinki.fi

Published in the Doctoral School in Natural Sciences Dissertation Series

Abstract

With the ongoing climate change economic activity in the Arctic steadily increases and it is expected to grow further in the coming years. However, harsh weather conditions of the present-day Arctic place strong demands for accurate and timely weather forecasts, which nowadays are obtained by means of numerical weather prediction. Sea ice covers a considerable part of the Arctic Ocean and numerical weather prediction systems operating in the region require a reliable and computationally-efficient representation of the sea ice cover in the model. Traditionally, simplified one-dimensional parameterisation schemes are applied for this task. However, implications of utilising such schemes in the context of contemporary high-resolution regional operational numerical weather prediction are not well studied. The present work aims to assess these effects through a series of numerical experiments in the operational-like environment. A new one-dimensional parameterisation scheme, allowing for varying level of complexity, implemented in the HARMONIE-AROME numerical weather prediction system, is used as the main research tool. The findings show that applying an over-simplified parameterisation scheme can result in considerable deterioration of the ice surface temperature field in the model. Errors in the modelled ice surface temperature influence the turbulent exchange between the ice surface and the model atmosphere, and, as a result, the near-surface atmospheric variables, such as the screen-level air temperature. Thus, improving the ice surface temperature in the model results in a positive impact on the atmospheric forecast of these parameters. Therefore, a sea ice scheme within an operational numerical weather prediction system should preferably include an explicit representation of the snow layer to accurately represent the surface energy budget of sea ice. Applying a sea ice data assimilation procedure to assimilate a near real time satellite ice surface temperature product in HARMONIE-AROME further reduces the root mean square error of the ice surface temperature and improves the screen-level air temperature forecast over Svalbard and Franz Josef Land archipelagos, however the positive effect in the ice surface temperature is short-lived and greatly reduced already after three hours of model forecast. Spatial resolution of contemporary regional numerical weather prediction systems allows some of the fine-scale features of the sea ice cover to be explicitly represented in the model. Numerical experiments show that introducing irregular structures within the marginal ice zone of the model ice cover results in both local and non-local responses in the atmospheric model. These findings indicate potential benefits of applying high-resolution ice cover in regional numerical weather prediction systems, for example, as a surface perturbation in ensemble prediction systems.

Acknowledgements

The present thesis is based on the research conducted at the Norwegian Meteorological Institute over a time period of more than five years. Indeed, it was a long journey and being a PhD student at the University of Helsinki while physically staying and working in Norway added a fair share of complications, sometimes expected and sometimes not. Just ten years ago, sitting in a train and preparing to spend a year in the military service, I would never have imagined that one day I will be working as a researcher abroad let alone pursuing a degree. But nevertheless, here I am, sitting in my office in Oslo and typing these words while realising that soon this pursuit will reach its conclusion, and I must say, I would not be able to reach this far without help and encouragement from many people.

First and foremost, I would like to thank Dr. EKATERINA KURZENEVA for encouraging me to enrol as a PhD student at the University of Helsinki, without her support this whole thing would never happen. Also, I am deeply grateful to my supervisor, Prof. HEIKKI JÄRVINEN for his endless patience and practical advice.

Research can not be performed by sitting alone in a locked room and I would like to thank my coauthors Dr. MARIKEN HOMLEID, Dr. MALTE MÜLLER and Dr. EKATERINA KURZENEVA, whom I already mentioned, for all their support and fruitful collaboration. To all the current and former colleagues at the Development Centre for Weather Forecasting and at the Norwegian Meteorological Institute in general, thank you for sharing your expertise and helping, especially in the beginning, with all the issues I encountered when running numerical experiments. Separately, I would like to thank the director of the Development Centre for Weather Forecasting Dr. JØRN KRISTIANSEN for finding ways to extend my position at the Norwegian Meteorological Institute and continue this project, and Dr. TERESA REMES for her friendly advice and encouragement. Special thanks to Dr. ANDREW SINGLETON and Dr. ERIN THOMAS for their numerous suggestions on improving the language of the manuscripts. Additionally, I would like to thank all the colleagues outside the Norwegian Meteorological Institute, especially at the Swedish Meteorological and Hydrological Institute and at the Finnish Meteorological Institute, for their support and willingness. There are too many names to list all of them here, but in no case I deem your contributions and support less important.

I would like to thank the pre-examiners Res. Prof. DAVID BROMWICH from the Ohio State University and Res. Prof. JARI HAAPALA from the Finnish Meteorological Institute for finding time to evaluate the thesis. Also, I am grateful to Dr. STEFFEN TIETSCHE from the European Centre for Medium-Range Weather Forecasts for agreeing to act as an opponent at the defence.

Finally, I would like to emphasise the importance of financial support provided to this PhD project throughout the years. Research presented in this thesis was funded through various external and internal projects at the Norwegian Meteorological Institute. Without the funding from such projects as ALERTNESS and the Nansen Legacy, to name a few, this study would not reach its current state.

10pul 5aspan Ocus, 14020 2022 roya

Acronyms and abbreviations

3DVAR	Three-dimensional VARiational analysis
ALADIN	Aire Limitée Adaptation dynamique Développement InterNational
AROME	Applications of Research to Operations at Mesoscale
CMEMS	Copernicus Marine Environment Monitoring Service
COSMO	Consortium for Small-scale Modeling
ECMWF	European Centre for Medium-Range Weather Forecasts
EKF	Extended Kalman Filter
ESTD	Standard deviation of errors
GFS	Global Forecast System
HARMONIE	HIRLAM-ALADIN Research on Mesoscale Operational NWP in Eu-
	romed
IFS	Integrated Forecasting System
LSMIX	Large-scale mixing
ME	Mean Error
MODIS	Moderate Resolution Imaging Spectroradiometer
NCEP	National Centers for Environmental Prediction
NPP	National Polar-orbiting Partnership
NWP	Numerical Weather Prediction
RMSE	Root Mean Square Error
SICE	Simple ICE scheme
SURFEX	SURFace EXternalisée
VIIRS	Visible Infrared Imaging Radiometer Suite

Contents

Lis	ist of original publications									
1	Introduction									
2	Methods	17								
	2.1 Numerical models	17								
	2.2 Observational data	19								
	2.3 Model data	20								
	2.4 Design of numerical experiments	21								
3	Results	25								
	3.1 Effects of applying a sea ice model in a short-range regional NWP system	25								
	3.2 Importance of the evolving ice thickness and snow cover	28								
	3.3 Necessity of constraining the ice state in the model	31								
	3.4 Atmospheric response to small-scale features in the sea-ice cover	34								
4	Discussion									
5	5 Conclusions									
6	6 Summary of the original publications									
Bil	3ibliography 4									

List of original publications

This thesis consists of an introductory review, followed by four peer-reviewed research articles. Papers **I–III** are reprinted under the terms of the Creative Commons Attribution 4.0 International license, and paper **IV** is reprinted under the terms of the Creative Commons Attribution Non Commercial No Derivatives 4.0 International license.

- I. Batrak, Y., Kourzeneva, E. & Homleid, M. Implementation of a simple thermodynamic sea ice scheme, SICE version 1.0-38h1, within the ALADIN-HIRLAM numerical weather prediction system version 38h1. *Geoscientific Model Development* 11, 3347– 3368. DOI: 10.5194/gmd-11-3347-2018 (2018).
- II. Batrak, Y. & Müller, M. On the warm bias in atmospheric reanalyses induced by the missing snow over Arctic sea-ice. *Nature Communications* 10, 4170. ISSN: 2041–1723. DOI: 10.1038/s41467-019-11975-3 (Sept. 2019).
- III. Batrak, Y. Implementation of an Adaptive Bias-Aware Extended Kalman Filter for Sea-Ice Data Assimilation in the HARMONIE-AROME Numerical Weather Prediction System. *Journal of Advances in Modeling Earth Systems* 13, e2021MS002533. DOI: 10.1029/2021MS002533 (2021).
- IV. Batrak, Y. & Müller, M. Atmospheric Response to Kilometer-Scale Changes in Sea Ice Concentration Within the Marginal Ice Zone. *Geophysical Research Letters* 45, 6702– 6709. DOI: 10.1029/2018GL078295 (2018).

Chapter 1 Introduction



EA ICE, these two short words are strongly associated with one of the harshest environments in the Northern Hemisphere, the Arctic. Indeed, even though the

seasonal ice cover reaches as far as $40^{\circ}N_{2}^{I,2}$ the vast area of the Arctic seas encrusted by seasonal and perennial ice outweighs the seasonally frozen mid-latitude seas by an order of magnitude. These frozen seas are covered by myriads of moving ice floes of all different shapes and sizes forming endless planes scarred by chaotic ridges or swiftly diverging in their constant motion creating narrow openings and wide polynyas. Sea ice, especially when covered by snow, has low thermal conductivity and high albedo insulating the ocean from the atmosphere and reducing the amount of absorbed shortwave radiation. All this creates a unique environment affecting the life on Earth on all scales: from providing a stable platform for various microorganisms; primary producers³ and animals³ to moderating the climate⁴⁻⁶ over millennia?

Earth climate and Arctic sea ice are strongly connected, with ice influencing even the areas located thousands kilometres away from the polar regions. However, sea ice impacts not only the long-term evolution of the climate but also has a strong influence on the shortterm atmospheric state - weather. Drastic difference in the physical properties of open water and sea ice results in a strong surface temperature gradient between ice-free and ice-covered parts of the Arctic Ocean and affects the turbulent energy exchange between the ocean surface and the atmosphere. This in turn has impact on weather not only over ice-covered regions but in remote areas as well. A simple example: marine cold air outbreaks or advection of cold air masses formed over ice-covered areas towards warm ice-free ocean which can result in formation of Arctic fronts and polar lows.8,9

Severe weather events can have disastrous consequences and socio-economic activities in any region on Earth require timely and accurate information about current and expected weather. Nowadays this information is provided by means of numerical models approximating the real processes on Earth surface and in the atmosphere. There is a wide range of NWP models used for operational weather prediction but fundamentally they can be divided in two classes: global and regional. The model domain of a global model covers the whole Earth Earth and has an important advantage of not having discontinuities or boundaries in the model atmosphere, except for the surface and top of the atmosphere, thus such models do not require external forcing data to account for the unrepresented part of the atmosphere. Examples of such models are IFS by the ECMWF and GFS by NCEP. However, advantages of global grids lead to a serious limitation of such models - high computational and storage cost. Computational cost of global models is often a limiting factor and to alleviate it operational centres, which require timely weather forecasts, commonly use models with reduced spatial resolution and simplified representation of the atmospheric processes. From the other side, regional NWP systems do not cover the whole globe but rather provide weather forecast for a limited area. Thus, such systems require information from a global model or another regional model covering a larger region to define the atmospheric state over the model domain boundaries. Although, since these systems cover smaller areas, they are less computationally demanding and operational centres can afford using higher spatial resolution or more advanced representation of physical processes, for example, applying non-hydrostatic formulations.

Operational NWP systems routinely provide forecasts for any region on Earth, although some regions are proven to be more challenging for weather prediction than others. One of such regions is the Arctic, where lack of in-situ observations combined with simplifications applied in the weather models, usually tuned to perform best in the mid-latitudes, results in reduced forecast skill!^o Quality of weather forecasts for the Arctic regions becomes more and more pressing issue with the recent climate change and increasing activity in the region.^{II} The Arctic, which remained a deserted and isolated place for many centuries, experiences a more drastic warming than any other region on Earth¹²⁻¹⁴ For a long time the possible advantages of sea routes by the Arctic Ocean were only theorised upon^{15,16} and many brave men lost their lives trying to discover the North-West and North-East passages¹⁷⁻¹⁹ In the first quarter of the XXI century, with retreating ice cover and increasing ship traffic, importance of the Arctic Ocean to the world economy grows and it is predicted to increase in the following decades²⁰ Growing activity in the region results in increasing demands on the quality of weather forecasts.

In NWP systems representation of surface processes plays an important role impacting the quality of the produced forecasts, and surface parameterisation packages become more and more advanced. Naturally, Earth surface forms the lower boundary condition for the model atmosphere where turbulent and radiative fluxes transfer energy between the surface and the atmosphere. All these fluxes should be accurately represented or parameterised in an NWP model to give a reliable forecast of nearsurface parameters, such as screen-level air temperature (here screen-level denotes the air temperature modelled or observed at the height of a standard meteorological screen, which is situated at the 1.25 to 2 m height above ground²¹). A surface parameterisation package of an NWP system usually includes representation of various surface types including natural landscapes, urban areas and water bodies. Additionally, if an NWP system is designed to operate in high latitudes, its surface parameterisation package should include a numerical representation of the sea ice cover – a sea ice model.

Numerical modelling of sea ice went a long way from first semi-empirical formulations to contemporary

contemporary numerical models. Early studies focused mainly on the evolution of the ice thickness as a response to the atmospheric conditions^{22,23} to derive a practical means of predicting the ice thickness from the air temperature series. However, these simple models, while being undoubtedly useful to practical men, do not resolve the evolution of the ice cover with enough level of detail for incorporating them in NWP systems. Following evolution of numerical sea ice models went along the two main directions, namely: thermodynamic and dynamic ice modelling. Thermodynamic sea ice models aim on representing the processes in ice column with maximum possible level of detail while dynamic models focus on the larger-scale evolution of the ice cover as a drift medium. Finally, these two types of sea ice models are merged into dynamicthermodynamic models trying to accurately represent the full evolution of the sea ice cover. These models can use somewhat simplified representations of the thermodynamic processes to reduce the computational costs emerging when running the model on the basin scales.

Traditionally in numerical forecasting, NWP systems and ocean models are operated separately and represent sea ice cover with different level of detail. In ocean modelling one of the main products is the sea ice forecast which necessitates an advanced dynamicthermodynamic sea ice model to be included. These forecasts focus mainly on such parameters as ice thickness and ice compactness while thermal properties of the ice cover are of less importance. From the other side, in NWP sea ice is often treated simply as the lower boundary condition of the atmospheric model and not as one of the main components of the system, therefore in these systems surface properties of the sea ice cover become the most important for a parameterisation scheme to accurately resolve since these parameters influence the energy exchange between surface and the atmosphere. As a result, sea ice models in NWP systems are often greatly simplified, sometimes to the point of prescribing the ice temperature by means of an external data set or climatology.²⁴

Additionally, the so-called fully coupled NWP systems are in active development and become more common in operational applications^{25–27} These systems, in case of ocean–sea-ice–atmosphere coupling, include both an atmospheric and a three-dimensional ocean model with a dynamic-thermodynamic sea ice model allowing for a great detail of representing the interactions between these components. However, combining multiple models in a single system not only considerably increases the computational cost of model simulation, but also introduces additional degrees of freedom in the system making model tuning more complicated.

Contemporary short-range NWP systems operating over the Arctic model domains use various approaches to representing the sea ice cover. Even though prescribing ice temperatures in the model is becoming obsolete nowadays, it is still a relatively popular solution in the systems which are not focused on the high latitudes^{24,28}In models with prognostic sea ice schemes, complexity and design of the applied parameterisation also varies. Usual choices are single layer and multilayer schemes with prescribed ice thickness, snow layer is often missing or represented by modifying the albedo of the ice cover. However, whilst modern NWP systems use finer and finer resolution grids, with the grid cell dimensions nearing 1 km, and become more and more advanced in general, effects and limitations posed by traditionally-applied

traditionally-applied simplistic sea ice parameterisation schemes are not well studied.

The aim of the present work is to develop a sea ice parameterisation scheme applicable to regional short-range high-resolution operational NWP and improve the understanding of the effects induced by simplified parameterisation schemes on model forecasts. Specifically, the following research questions are addressed throughout the thesis:

- What are the effects induced by applying a thermodynamic sea ice scheme into a short-range regional NWP system operating in the Arctic? Ice surface temperature in the model can be prescribed by means of an external observational or modelling data set, which potentially could have a higher quality than the field computed by a simplified parameterisation scheme. The aim is to investigate the benefits of applying a prognostic scheme in the model and assess whether a simplified parameterisation scheme is always a viable option compared to prescribing ice surface temperature in the system.
- What are the consequences resulting from applying a simplified representation of the evolution of the sea ice cover and their impact on the near-surface atmospheric variables in the model forecast? NWP systems apply a wide set of simplifications in their sea ice parameterisation schemes, such as prescribed ice thickness or binary ice cover, while some of them can be detrimental to the quality of the model forecast.

- Should the state of the sea ice parameterisation scheme in an NWP system be constrained by observational data sets? Ice surface temperature is seldomly constrained by the observations in operational NWP applications. However, applying a data assimilation procedure could potentially alleviate the modelling errors caused by the simplicity of the parameterisation scheme.
- Are there potential benefits of applying high-resolution representation of the sea ice cover in a high-resolution convectionpermitting NWP system? Contemporary regional NWP systems typically define sea ice cover by means of low-resolution satellite or modelling products. Impacts of applying high-resolution sea ice cover on the evolution of the atmospheric variables are not well-studied in the context of operational NWP.

This thesis is organised as follows: Chapter 2 provides an overview of the used NWP system and applied sea ice parameterisation scheme, utilised external observational and modelling data sets, as well as design of the performed numerical experiments. Chapter 3 presents a summary of the main results, Chapter 4 discusses the obtained results and Chapter 5 concludes the thesis and highlights the prospects for future research. The introductory review is followed by the four reprinted articles, prefaced by a short summary of the published peerreviewed articles with a description of author's contributions in case of joint publications.

Chapter 2

Methods

This chapter provides an overview of the applied numerical models and observational data sets. Additionally, a summary of the numerical experiments performed in the present work is given.

2.1 Numerical models

In this thesis the HARMONIE-AROME²⁹ NWP system is used as the main research tool for studying the effects of applying a sea ice parameterisation scheme in regional NWP systems. HARMONIE-AROME includes a regional non-hydrostatic convection-permitting atmospheric model and a selection of atmospheric and surface data assimilation procedures. Typical operational configurations use 3DVAR for atmospheric data assimilation and optimal interpolation for surface analysis. Surface processes in HARMONIE-AROME are represented by the externalised modelling platform SUR-FEX³⁰ which provides parameterisation of various land surface types, such as inland water bodies, seas, urban areas, and natural landscapes. SURFEX is incorporated as a subcomponent of HARMONIE-AROME and resolves the energy exchange between the atmosphere and the underlying medium. In the present thesis HARMONIE-AROME is configured to run on a relatively dense grid with spatial resolution of 2.5 km and 65 vertical layers; the top-most model layer is located at the 10 hPa height, and the lowest model layer has the height of approximately 12.5 m above the model surface. For the model integration HAR-MONIE-AROME uses a semi-implicit numerical scheme which is combined with a semi-Lagrangian advection scheme, and a characteristic time step for a 2.5 km grid is 60 s.

Numerical experiments in this thesis are designed to resemble operational applications of HARMONIE-AROME where model integration is performed by applying the so-called cycling approach. With this approach, model output from a numerical experiment represents a set of overlapping model forecasts rather than continuous series of model states, which is usually the case in climate modelling, for example. Schematic on the Figure 2.1 provides details on the organisation of a typical HAR-MONIE-AROME experiment in NWP mode.

HARMONIE-AROME is a regional NWP system, therefore it requires boundary conditions to be provided from an NWP system covering a larger area. In the present work HARMONIE-AROME is forced by the global NWP system IFS-HRES operated by the ECMWF. Boundary conditions are provided with one hour temporal resolution.

Originally,



Figure 2.1: Schematic representation of running a HARMONIE-AROME model simulation in NWP mode. Model experiment consists of a series of analyses and model forecasts or *cycles*, where model forecast from the previous cycle is provided as the initial model state for the next cycle's analysis. Main cycles (oo and 12 UTC on the schematic) provide longer model forecasts, while forecast length at intermediate cycles is only enough to obtain the initial model state for the next analysis. Also on the schematic, a simplified representation of the cycling strategy of the host model, IFS-HRES, is outlined.

Originally, sea-ice covered areas in HAR-MONIE-AROME were represented in a very simplified way: the ice surface temperature, as well as the sea surface temperature, was prescribed by means of an external data set and remained constant during model forecast. To represent the effects of the evolution of the sea ice cover in the model, the prescribed ice surface temperature field was updated each cycle during the analysis step. Paper I introduces a new thermodynamic sea ice parameterisation scheme, SICE, in HARMONIE-AROME to resolve the evolution of the ice surface temperature during the model forecast. The scheme resolves the process of heat diffusion in the ice layer with constant salinity of 3 ‰. Additionally, the scheme allows for the snow cover over sea ice to be explicitly represented by an

adapted version of one of the land snow parameterisation schemes provided by SURFEX³¹ However, the process of snow-ice formation, which is of less importance in the Arctic than in the Antarctic³² although indicated to become a more common phenomenon in coming years³²⁻³⁴ is not represented in the current version of the scheme. Turbulent fluxes over a partially ice-covered sea grid cell are computed as a weighted mean between sea-ice and open water fluxes with a discretionary form drag term.³⁵ Original implementation of SICE uses prescribed and constant ice thickness, and Pa**per II** extends this implementation by adding the processes of ice growth, and ice melting from bottom and surface. The ice growth formulations in SICE use a prescribed constant ocean heat flux as a tuning parameter.

The

The SICE scheme does not resolve ice dynamics, and it is unable to freeze new ice from open water or melt the ice completely since NWP applications of SURFEX in HAR-MONIE-AROME use a prescribed sea surface temperature field in place of a prognostic parameterisation for sea grid cells. Thus, sea ice cover in HARMONIE-AROME is defined by the sea ice concentration field provided to the model from an external source, for example from the host model. The ice concentration field is updated each cycle at the analysis step and remains constant during the model forecast. Such approach, while allowing a considerable simplification of the sea ice component of HARMONIE-AROME, introduces inconsistencies between the existing and updated ice cover in the model. For initialisation of the new ice grid cells SICE applies a simple distanceweighted extrapolation procedure using the ice state from nearest grid cells with existing ice cover. If the model state from the previous model forecast has no ice-covered grid cells, new ice is initialised as a snow-free surface with predefined uniform ice thickness and temperature equal to the freezing temperature of the sea water.

The sea ice parameterisation scheme discussed in **Papers I, II, IV** of this thesis does not use any additional information to constrain the state of the sea ice model, thus ice temperature profile, as well as ice thickness and snow cover over sea ice (when applicable), evolve freely during the model integration and from cycle to cycle. **Paper III** further extends the SICE model by adding a data assimilation procedure correcting the thermal profile of the ice slab. This assimilation procedure uses the EKF to assimilate a near real time ice surface temperature product³⁶ which provides estimated ice surface temperature only in cloudfree conditions. Therefore, before performing sea ice analysis within SURFEX, which operates with one-dimensional parameterisations, ice surface temperature product should be converted to gridded observations defined on the HARMONIE-AROME model grid. This step is preformed by applying the optimal interpolation procedure, which uses the surface temperature field from the previous cycle's forecast as the model background. For the ice-covered grid cells located at the considerable distance from the valid ice surface temperature product pixels the system reports missing observations and the analysis procedure does not update the SICE model fields, only propagating the background error covariance matrix.

2.2 Observational data

In this thesis various observations are used for studying and verifying the performance of the sea ice parameterisation in the HAR-MONIE-AROME NWP system, and for assessing its effects on the other components of the system.

The primary effects of applying the sea ice parameterisation scheme are observed in the ice surface temperature fields, and for verifying the performance of the scheme observational data sets of the ice surface temperature are required. However, in-situ observations are scarce and often report only local conditions³⁷ or do not directly measure surface temperature, making them less applicable for studying general effects of introducing a new parameterisation scheme. Thus, in this thesis a considerable amount of satellite remote-sensing product data is used for verification purposes. Given that ice surface temperature is a fastevolving variable^{37,38} daily aggregated ice surface temperature fields are not optimal for verification purposes since they combine multiple

ple satellite swathes with different valid times, which would make the analysis considerably more difficult. Instead, in the present thesis L2³⁹ near real time products are utilised allowing for better temporal collocation of model and observational data.

The following satellite ice surface temperature products are used for verification purposes in **Paper I** and **Paper III**: MOD29⁴⁰ and MYD29⁴¹ derived from the data obtained by the MODIS instrument on-board the Terra and Aqua satellites, respectively. Additionally, **Paper I** uses the VNP30⁴² product obtained from the VIIRS instrument on-board the Suomi NPP satellite for extended validation, and **Paper II** uses an L4³⁹ ice surface temperature product from CMEMS⁴³ for assessing the ice surface temperature of atmospheric reanalysis products in cloud-free conditions.

In-situ observations on the sea ice, though scarce, are a valuable source of local information which can not be obtained from spaceborne instruments. **Paper II** of the present thesis uses observations from the N-ICE2015 drift campaign⁴⁴ for assessing the ability of different sea ice parameterisation schemes to represent evolution of the ice surface temperature with focus on clear-sky events. Ice surface temperature along the N-ICE2015 drift trajectory is derived from the in-situ longwave radiation observations.

In-situ observations and remote sensing data sets providing information about the icecovered regions are used to verify the direct effects of applying a sea ice parameterisation scheme as a component of an NWP system. However, the indirect effects, such as impact on the screen-level air temperature forecast in the surrounding regions, are also of high importance, especially since they represent the forecast quality over the populated areas (for example, the Svalbard archipelago with total population close to 2500 individuals as of 2021). Therefore, **Paper I** and **Paper III** use observations from the land meteorological (SYNOP) stations, which routinely report to the Global Telecommunication System, to verify the forecast quality of the screen-level air temperature and humidity, and 10 m wind speed.

State of the sea ice cover directly affects the energy exchange between the surface and the atmosphere thus modifying the state of the near-surface air masses, which in turn can have an impact on the evolution of the upper air variables. To assess these possible effects, accurate information about the vertical structure of the atmosphere is required, which is usually obtained by means of radio sondes. For example, **Paper III** uses soundings from the Ny Ålesund station located at the western coast of the Spitsbergen island to study the possible effects of different initial states of the sea ice cover on the atmospheric forecast. However, since there are only a few operational sounding stations in the European Arctic^{45,46} and all of them are located over land, which limits the applicability of such observations, this type of observations is used only for the preliminary assessment of the induced effects. A more detailed study on the responses induced by the sea ice model in the upper levels of the atmospheric model would require considerably longer than discussed in the present work experimental periods for generating reliable statistics.

2.3 Model data

Different atmospheric modelling systems use different approaches to parameterising sea ice cover. Thus, comparing the products generated by various NWP systems against a reference ence data set allows to estimate the quality of the represented evolution of the sea ice state with respect to applied sea ice parameterisation schemes. However, operational NWP systems are constantly evolving and do not provide long series of model forecasts produced with the same version of the model system. Additionally, timeliness constraints of an operational system pose strict limitations on the scheduled start time of a model cycle and operational systems routinely use only observations which arrive before the cut-off time in the analysis⁴⁷ even though additional observations could become available later. To alleviate these limitations of operational NWP, dedicated non-real-time reanalysis48 products are produced. An atmospheric reanalysis product is generated with a single version of an NWP system which is not modified throughout the whole time period covered by the product. Moreover, since reanalysis products are always generated for periods in past and not in real time they can use a much wider set of observations than operational systems. Thus, these products are often used as the best estimate of the surface and atmospheric state⁴⁹ for a specific date.

In this thesis, a limited set of contemporary reanalysis products is used to study the effects induced by simplifying assumptions applied in the sea ice parameterisation schemes. **Paper II** uses the following global atmospheric reanalysis products: MERRA-2,⁵⁰ ERA-Interim,⁵¹ ERA5,⁵² JRA-55,⁵³ and NCEP-DOE Reanalysis 2.⁵⁴

2.4 Design of numerical experiments

In the present thesis effects induced by applying a sea ice parameterisation scheme in HARMONIE-AROME are studied through a series of numerical experiments and simulations. The experiments are configured to follow the design of the operational applications of HARMONIE-AROME and use model domains, which include considerable amount of ice-covered regions. A general overview of the model domains and conducted numerical experiments discussed in the present work is provided by Figure 2.2 and Table 2.1. On the Figure 2.2 AA denotes the AROME-Arctic domain which is used for operational weather forecasting⁵⁵ by the Norwegian Meteorological Institute. On the same figure, AM denotes the AROME-MetCOOP domain, which was used for operational weather forecasting jointly⁵⁶ by the Norwegian Meteorological Institute and the swedish Meteorological and Hydrological Institute. The AROME-MetCOOP domain was extended multiple times over the years, the present work uses one of the earliest domain configurations which was in operational service before September 2017. The large size of the AA and AM model domains results in high computational and storage costs of running numerical experiments on these domains. Thus, to allow performing model simulations over longer time periods without using prohibitively expensive operational model configurations, a smaller model domain is used in a number of experiments presented in this thesis. This model domain, marked as AC on the Figure 2.2, includes a considerable amount of sea-ice covered areas, seasonal and perennial, as well as both Svalbard and Franz Josef Land archipelagos allowing for validation of model results against observations from the coastal SYNOP stations.

Operational NWP systems run on the AA and AM domains use data assimilation procedures in both the surface parameterisation package and the atmospheric model to constrain the model state by the observations. As of 2022 these systems use 3DVAR for atmospheric



Figure 2.2: General overview of the study area and available in-situ observations. The map shows median March and September sea ice extent in the Arctic seas computed for the period from 2010 to 2020. Also on the figure, the positions of various model domains used for numerical experiments throughout this thesis are outlined. A summary of the land SYNOP stations located in the vicinity of the ice-covered areas, which were used for verifying the performance of HARMONIE-AROME is provided in the insets. Additionally, the top inset shows the drift trajectories of the four legs of the NICE-2015 drift campaign.

Table 2.1: Overview of the performed numerical experiments.	Notes	· Binary ice cover, prescribed ice temperature		Prognostic ice temperature		As \mathbf{I}_{α}^{AA} but with snow cover added	Climatological ice thickness	Form drag is applied		Spin-up period until 10 January 2015		Assimilation of a sea ice temperature product	Initialised from \mathbf{m}_{α}^{Ac} state on 30 November 2019	Based on \mathbf{H}_{α}^{AC} same initialisation as in \mathbf{H}_{REF2}^{AC}	As III s ^c but with surface analysis used in 3DVAR		Modified ice cover in the marginal ice zone		the number of snow lavers and denominator denotes the		\cdot ice thickness computation/initialisation method ($clim$ –	formulations of the SICE scheme).
	Time period	1 Mar 2013 – 30 Apr 2013	1 Mar 2013 – 31 Mar 2013	1 Mar 2013 – 31 Mar 2013	1 Mar 2013 – 30 Apr 2013	1 Mar 2013 – 30 Apr 2013	1 Mar 2013 – 30 Apr 2013	1 Mar 2013 – 31 Mar 2013	10 Jan 2015 – 20 Mar 2015	1 Sep 2014 – 20 Mar 2015	1 Sep 2019 – 31 Jan 2020	1 Sep 2019 – 31 Jan 2020	1 Dec 2019 – 31 Dec 2020	1 Dec 2019 – 31 Dec 2020	1 Dec 2019 – 31 Dec 2020	10 Jan 2015 – 30 Apr 2015	10 Jan 2015 – 30 Apr 2015		provided on the Figure 2.2. Where numerator denotes		ed uniform ice thickness, or	2g – prognostic ice thickness
	UA^{d}	3DVAR	3DVAR	3DVAR	3DVAR	3DVAR	3DVAR	3DVAR	١	١	LSMIX	LSMIX	3DVAR	3DVAR	3DVAR	١	۰		efinitions		of prescrib	ckness; <i>pn</i>
	Ice thk c	۲	١	o.75	o.75	o.75	clim	o.75	o.75	prog	prog	prog	prog	prog	prog	o.75	o.75	ain, following the de on of the SICE schen	wing the de sice scher		es, in case c	ical ice thic
	SICE^{b}	١	١	%	%	3/4	%	%	%	3/4	12/4	12/4	12/4	12/4	12/4	%	%		ice layers.	s in metro	dimatolog	
	omain ^a	AA	AM	AM	$\mathbf{A}\mathbf{A}$	$\mathbf{A}\mathbf{A}$	$\mathbf{A}\mathbf{A}$	$\mathbf{A}\mathbf{A}$	AA	AC	AC	AC	AC	AC	AC	AA	AA		odel dom nnfigurari	umber of i	e thickne:	escribed (
	L	$I_{\rm R EF}^{\rm AA}$	I REF	\mathbf{I}_{α}^{AM}	$\mathbf{I}_{\alpha}^{\mathbf{AA}}$	দ্ব	Ιγ	I	II ^{AA} REF	$\mathbf{II}_{\alpha}^{\mathbf{AC}}$	III ^{AC} REFI	$\mathbf{III}_{\alpha}^{AC}$	III ^{AC} REF2	III BC		IVAREF	$\mathbf{IV}_{\alpha}^{\mathbf{AA}}$		$_{p}^{a}$ W	n d	6) Ic	pr

^d) Atmospheric data assimilation/initialisation procedure, missing value denotes a free-running atmospheric model where HAR-MONIE-AROME is forced only on the domain boundaries.

spheric data assimilation and optimal interpolation for surface analysis^{55,56} However, for numerical experiments and sensitivity studies applying computationally expensive 3DVAR is less beneficial, or even detrimental in situations where discontinuities introduced by the data assimilation procedure are undesirable. Thus, in the present thesis atmospheric data assimilation is applied only in the experiments $\mathbf{I}_{\text{REF}}^{AA,AM}$ and $\mathbf{I}_{\alpha-\delta}^{AA,AM}$ assessing possible effects of applying a sea ice parameterisation scheme in the operational NWP environment, and in the experiments $III_{REF2,\beta,\gamma}^{AC}$ studying the effects of sea ice analysis on the atmospheric data assimilation. Numerical experiments discussed in the Paper II and Paper IV were conducted without applying atmospheric data assimilation and these studies assess continuous series of weather forecasts generated by HAR-MONIE-AROME. Experiments \mathbf{III}_{refi}^{AC} and $\mathbf{III}_{\alpha}^{AC}$ use large-scale mixing to combine the model state of HARMONIE-AROME with atmospheric fields provided by the host model as the initialisation procedure for the atmospheric model. This approach allows saving computational resources in these considerably long experiments while still benefiting from the corrected initial state of the model atmosphere?7

When prognostic formulations for the sea ice thickness are applied in the SICE scheme, as in \mathbf{u}_{α}^{AC} and experiments $\mathbf{III}_{\alpha\gamma}^{AC}$ discussed in **Paper III**, the scheme is initialised with uniform ice thickness of 0.75 m and ice temperature equal to the freezing point of the sea water (assuming the water salinity of 35 %). If SICE is configured to explicitly model the snow layer on top of the ice, HARMONIE-AROME starts from the snow-free state and SICE accumulates snow cover from model precipitations throughout the experimental period.

All the numerical experiments presented in this thesis, except the experiments discussed in the **Paper III** devoted to the problem of sea ice data assimilation, use the standard surface data assimilation procedure of HARMONIE-AROME which consists of a two-dimensional optimal interpolation procedure generating gridded observations¹⁸ and a one-dimensional optimal interpolation procedure updating the prognostic fields in sURFEX. Additionally the surface analysis procedure updates the sea surface temperature and ice concentration (if applicable) fields using the lower resolution fields provided by the host model.

Boundary conditions in the discussed model experiments are provided by the global NWP system IFS-HRES in a manner simulating the operational applications. For example, in operational environments a HAR-MONIE-AROME forecast with valid time of 00 UTC can not use IFS boundaries with the same valid time, since the corresponding cycle of the host model is not finished by the time when HARMONIE-AROME starts. Thus, operational applications HARMONIE-AROME use the most recent and complete IFS forecast when generating boundary conditions for a model cycle. In the conducted numerical experiments this behaviour is achieved by introducing an at least six hour delay between the valid time of IFS-HRES and HAR-MONIE-AROME forecasts, see the Figure 2.1 for the details.

Chapter 3

Results

3.1 Effects of applying a sea ice model in a short-range regional NWP system

Presence of the sea ice cover within the model domain strongly affects the energy exchange between the surface and the lowest model levels of the atmospheric model since sea ice, unlike open water, quickly responds to the atmospheric forcing signal, and has much higher albedo. This results in great variations of the ice surface temperature within the time frame of a single model forecast. As a consequence, prescribed ice surface temperature in HARMONIE-AROME can not adequately represent these processes leading to underestimation of the diurnal cycle of the near-surface air temperature over sea-ice covered regions. Additionally, for main cycles starting when the ice temperature is low, for example for the OO UTC cycle, prescribing the ice field by low nigh-time ice surface temperature leads to a cold bias in the screen-level air temperature forecast during the day-time in the model forecast. This situation is illustrated by the Figure 7a of the Paper I where prescribed ice surface temperature initialised at 00 UTC leads to a more than a 5 °C cold bias in the modelled screen-level air temperature after 12 hours of model forecast.

Introducing the SICE scheme in HAR-MONIE-AROME results in a responsive ice surface in the model, which freely evolves during the model forecast thus improving the diurnal cycle of the screen-level air temperature forecast over sea ice. This improvement is the most clear over the spring time in the Gulf of Bothnia where the diurnal cycle is prominent.

Another effect of applying a thermodynamic sea ice parameterisation scheme is the reduced error growth rate as a function of the forecast lead time. Figure 3.1 shows the performance of the SICE scheme in the operational application⁵⁵ of the HARMONIE-AROME NWP system over a one year period. As illustrated on the figure, in the reference configuration of HARMONIE-AROME the ice surface temperature RMSE computed against a near real time ice surface temperature product quickly grows with the increasing forecast lead time for all months, except the April-August period, when ice temperature is close to the melting point. Applying a simple thermodynamic sea ice scheme helps to reduce the error growth rates in the modelled ice surface temperature without deteriorating the model forecast. However, with a prescribed ice surface temperature field, HARMONIE-AROME shows smaller ice surface RMSE for short forecasts with lead time less than 12 hours. This can be explained by the fact that IFS-HRES, which is used



Figure 3.1: Ice surface temperature RMSE as a function of the forecast lead time computed over a one year period in two configurations of HARMONIE-AROME. On the figure, "HAR-MONIE-AROME without SICE" shows the simulated effect of prescribing the ice surface temperature in HARMONIE-AROME using the host model surface temperature field; "HAR-MONIE-AROME with SICE" represents the ice surface temperature in the operational system⁵⁵ which uses sea ice scheme in the same configuration as in I_{α}^{AA} . Scores are computed for the period from I December 2015 to I December 2016 for the forecast initialised at 00 UTC. On the panels, *x* axis: lead time in hours; *y* axis: RMSE in °C.



Figure 3.2: Relative change in the RMSE of the screen-level air temperature forecast resulting from applying a thermodynamic sea ice model in HARMONIE-AROME. RMSE is computed for up to 48 hour forecast lead time against in-situ observations.

used as a source of the prescribed ice surface temperature fields, while applying a similar to SICE sea ice parameterisation scheme with snow-free ice layer of prescribed and constant ice thickness⁹ uses the ice thickness of 1.5 m compared to 0.75 m in SICE. Having thicker ice cover, more suitable for a global NWP system, IFS-HRES reports lower ice surface temperature, which better corresponds to the actual temperature field, especially in the winter time.

Thus, for short-range NWP applications of

HARMONIE-AROME applying a high-quality prescribed ice surface temperature field would outperform a simplified thermodynamic sea ice parameterisation scheme, but for model forecasts longer than 12 hours even a simplified sea ice model has a considerable positive effect. This result is in line with the findings of a similar study discussing the effects of applying a thermodynamic sea ice model instead of using prescribed ice temperature in the COSMO model.²⁸

Effects of introducing a sea ice parameterisation scheme in HARMONIE-AROME can be traced also in the screen-level air temperature forecast for the stations located in the vicinity of the sea-ice covered areas. Verification scores computed against the SYNOP observations show that HARMONIE-AROME with SICE has lower air temperature RMSE than the reference model configuration for most of the stations located over the Svalbard archipelago and in the Gulf of Bothnia, as shown on the Figure 3.2. Other stations located within the model domain show a relatively weak response to introducing a thermodynamic sea ice model. However, there is a small number of Svalbard stations that show a considerable increase in the air temperature RMSE in the updated version of HARMONIE-AROME. This deterioration of the forecast quality is caused by the low resolution sea ice cover field provided by IFS-HRES, which is used to define sea-ice covered grid cells when SICE is used in HAR-MONIE-AROME. This low resolution ice concentration field, which uses a corresponding coarse land-sea mask, when interpolated to the 2.5 km grid of HARMONIE-AROME is unable to represent all the features of the coastal sea ice cover around Svalbard, such as fjords covered by the land-fast ice. As a result, in sensitivity experiments these fjords are ice-free, which causes

causes the warm bias in the modelled screenlevel air temperature. The reference configuration of HARMONIE-AROME uses the surface temperature field produced by IFS-HRES to derive the prescribed ice surface temperature, thus in the areas with complex coast line of the Svalbard archipelago the resulting 2.5 km ice temperature field is interpolated from snowcovered land grid cells of IFS-HRES therefore showing no spurious warm bias.

Additionally, applying fractional ice cover in HARMONIE-AROME can result in deteriorated 10 m wind speed forecast. For example, experiments \mathbf{I}_{α}^{AM} , \mathbf{I}_{α}^{AA} , \mathbf{I}_{β}^{A} and \mathbf{I}_{γ}^{AA} show a 0.4 ms⁻¹ increase in the mean error of the wind speed forecast over the coastal SYNOP stations. This effect is caused by introducing a contribution of the low-roughness open-water part of an ice-covered grid cell to the average turbulent flux of momentum, which is present in case of ice concentration less than 100%. Applying the form drag term when computing the average drag coefficient over a sea-ice covered grid cell (experiment \mathbf{I}_{δ}^{AA}) alleviates the observed increase in the wind speed bias.

3.2 Importance of the evolving ice thickness and snow cover

So far the effects of applying a sea ice parameterisation scheme in HARMONIE-AROME were tested with using the simplified SICE configuration which has prescribed and uniform ice thickness without any snow cover. However, these simplifications greatly reduce the accuracy of the represented evolution of the sea ice cover in the system. As a consequence, SICE in this configuration tends to show a noticeable warm bias in the modelled ice surface temperature, which is apparent when comparing the model forecast with satellite data, as presented on the Figure 3.I.





A popular approach to mitigating this limitation of the scheme is replacing the uniform prescribed ice thickness with non-uniform ice thickness;^{60,61} for example, based on a climatological data set as in the experiment \mathbf{I}_{γ}^{AA} . This varying in space prescribed ice thickness helps to somewhat reduce the warm bias in the model by the cost of introducing an additional external data dependency. However, applying a climatological ice thickness product still limits the ability of the system to adapt the ice cover state following the signals in the atmospheric forcing. Therefore, for operational applications a more preferable solution is extending the formulations of the SICE scheme to represent the processes of ice growth and melting. With this approach the ice scheme is able to represent the evolution of the ice thickness throughout the year while staying in balance with the state of the model atmosphere. Although, introducing a freely-evolving ice thickness in the system can lead to its unrealistic values, since it is not constrained by observations and a one-dimensional scheme can not represent sea ice as a drift medium.

Accurate and evolving ice thickness in the model, while being an important parameter, can not completely compensate other simplifications applied in the original configuration of the SICE scheme, most importantly the absence of the snow cover. Ice is almost always is covered by a layer of snow, which has different thermal properties and effectively insulates ice and underlying ocean from the atmosphere.^{62,63} Thus, when snow cover is present, surface temperature can be lower than in cases of snow-free ice surface. Without the snow cover ice in the model should be unrealistically thick to allow such low surface temperatures but still unable to resolve extremely fast changes in the surface temperature resulting

from the low thermal conductivity of the topmost layer of snow.

The original configuration of the SICE scheme discussed in the Paper I already provides an option to include a parameterisation of the snow cover in the system, however, initial tests showed that for the Svalbard stations applying the sea ice parameterisation with explicit representation of the snow cover does not result in clear improvement of the verification scores. This lack of clear positive impact, especially in the screen-level air temperature forecast, can be partly attributed to the configuration of the sensitivity experiment, which was performed over a two-month period in spring 2013 starting from the snow-free initial sate of the ice cover. Thus, snow cover accumulated from the model precipitations over the experimental period was considerably thinner than the snow cover in reality, which is close to its maximum thickness in the spring time before the snow melt starts. Nevertheless, comparison against the satellite ice surface temperature products shows that including the snow layer on top of the ice in the experiment \mathbf{I}_{β}^{AA} reduces the warm bias of the ice surface temperature forecast in HARMONIE-AROME (see, for example, Figure 9a in Paper I).

Further assessment of the effects of misrepresenting the sea ice state in simplified parameterisation schemes is performed by comparing different atmospheric reanalysis products. Results show that a parameterisation scheme with no snow layer and a prescribed and uniform ice thickness is unable to accurately represent the ice surface temperature field. As shown on the Figure 3.3 applying a parameterisation scheme with uniform ice thickness, especially when combined with binary ice cover, as in JRA-55 and NCEP-2, leads to strong biases over different parts of the Arctic ice cover. Notably,



Figure 3.4: Ice surface temperature in various reanalysis products and in HARMONIE-AROME experiments compared to the NICE-2015 drift campaign data. Considerable errors in the ice surface temperature are observed during the clear sky events, three such events are marked on the figure as CSE-1, CSE-2 and CSE-3.

tably, JRA-55 and NCEP-2, which have the thickest ice cover suitable for the central Arctic show considerable cold bias over areas covered by the thinner first-year ice. Including the snow cover in a sea ice parameterisation scheme considerably reduces the heat flux from ocean to the atmosphere. Additionally, insulating properties of the snow layer of any considerable thickness are so good that actual thickness of the ice layer beneath the snow plays only a minor role in the evolution of the ice surface temperature⁶³ thus rendering the model less sensitive to misrepresentation of the ice thickness field.

Effects induced by the explicit representation of the snow layer in a sea ice parameterisation scheme are the most apparent during the winter clear sky events when fast radiative cooling results in very low ice surface and, as a consequence, screen-level temperatures. Comparing the surface temperature fields of various atmospheric reanalysis products, as well as the HARMONIE-AROME forecasts from the experiments Π_{REF}^{AC} and Π_{α}^{AC} , against the in-situ observations from the drift campaign NICE- 2015 shows that all the model configurations which do not include the snow layer on top of the sea ice suffer from the 5 to 10°C warm bias. As shown on the Figure 3.4, the only model version which does not show this considerable bias during clear sky events is HAR-MONIE-AROME when uses SICE with explicitly represented snow cover. In the experiment \mathbf{H}_{REF}^{AA} SICE has no snow cover and uses the prescribed ice thickness of 0.75 m, which is thinnest amongst the compared products and results in the strongest warm bias during clear-sky events.

The findings presented in **Paper II** are in line with earlier studies⁶⁴ and indicate that contemporary atmospheric reanalysis products which are produced by models applying simplified representations of the sea ice cover tend to have warm bias of the ice surface temperature in certain conditions. Applying these data sets as a baseline or "best estimate" in model intercomparisons can lead to erroneous conclusions on the behaviour of compared systems. One example is the apparent "wintertime time cold bias" observed in climate models⁶⁵ found when comparing their surface energy budget with estimates based on warm-biased model reanalysis products. Additionally, as supported by the findings presented in the **Paper I**, this intercomparison shows that snowfree state of the ice cover is a strong simplification, which degrades the quality of the lower boundary condition for the model atmosphere.

3.3 Necessity of constraining the ice state in the model

In a typical NWP application the state of the sea ice cover, when modelled by the SICE scheme, freely evolves forced by the model atmosphere of HARMONIE-AROME. This allows for a reasonable representation of sea ice in the model without introducing additional data streams or time constraints in the operational environment. However, SICE does not represent all the processes of ice evolution and the ice surface temperature field in the model, while evincing a positive impact when compared to the original sea-ice handling approach of HARMONIE-AROME, shows considerable errors and biases when compared against observational data sets.

Paper III addresses this issue by applying a data assimilation procedure in SICE to constrain the sea ice state by an observational ice surface temperature product. Introduced framework uses a near real time ice surface temperature satellite product as observations, complemented by the ice concentration field provided by the host model. Numerical experiments performed over a small experimental domain show positive impact of assimilating ice surface temperature product in SICE. Specifically, the RMSE of the ice surface temperature forecast in the model experiment $\mathbf{III}_{\alpha}^{Ac}$ computed against an independent satellite ice surface temperature product is reduced by 0.4 °C on average after analysis, or in other words, at o hour forecast lead time. However, ice surface temperature is a very fast-evolving variable and effects of surface analysis found in the model forecast quickly diminish as a function of the forecast lead time. After 3 hours of model forecast the effect of sea ice data assimilation is already considerably reduced, but still traceable with ice surface temperature RMSE reduced by 0.1 °C on average compared to the reference experiment \mathbf{III}_{REFI}^{AC} without any sea ice data assimilation procedure applied.

Series of weekly model-domain-average ice surface temperature RMSE presented on the Figure 3.5 also show the generally positive effect of applying sea ice analysis in HAR-MONIE-AROME. RMSE values in both IIIREFI and $\mathbf{III}_{\alpha}^{AC}$ are highest at the beginning of the experimental period with values of 8.26 and 6.15°C, respectively. These high initial RMSE values are attributed to the effects of model cold start from the snow-free state of uniform ice thickness. By January 2020 these errors are reduced to 5.71 and 4.74°C, respectively. Assessing the time series of the RMSE components, namely ME and ESTD, (see the Figure 7 of the Paper III) shows that constraining the sea ice state by assimilating an observational data set considerably reduces the ice surface temperature ESTD, which indicates that applied procedure reduces the uncertainty of the ice surface temperature state in the system. Additionally, ESTD growth over a three hour model forecast in the $\mathbf{III}_{\alpha}^{AC}$ is higher than in the reference experiment III_{REFI}, which shows almost no change in the ESTD score. From the other side, the ME series of the ice surface temperature forecast does not show the same degree of improvement compared to the ESTD scores. Applying



Figure 3.5: Series of weekly ice surface temperature RMSE in HARMONIE-AROME with (the $\mathbf{HI}_{\alpha}^{AC}$ experiment) and without (the \mathbf{HI}_{REFI}^{AC} experiment) constraining sea ice parameterisation scheme by observations. RMSE scores are computed against an independent ice surface temperature product derived from MODIS data. On the figure, solid line represents the 0 hour forecast lead time; dotted line represents the 3 hour forecast lead time.

Applying the EKF-based sea ice analysis procedure was unable to completely remove the positive bias found in the ice surface temperature forecast of HARMONIE-AROME in certain conditions, although, this bias in the sensitivity experiment $\mathbf{III}_{\alpha}^{AC}$ is smaller than in the reference run III_{REFI}. This observed reduction of ME suggests that the applied bias-aware EKF is able to detect the presence of the model bias in the system. The apparent inability of the filter to completely remove the systematic error in the system can be attributed to several factors. Firstly, the strongest bias is observed at the beginning of the model experiments where crude initial conditions and scheme simplicity can not be fully compensated by adjusting the ice temperature. Secondly, incremental bias correction applied in SICE reduces the differences in the computed model bias at 0 and 3 hour forecast lead time. Finally, MODIS ice surface temperature products use infrared-sensitive channels of the instrument, thus they are available only

in cloud-free conditions and mismatch in the cloud mask between HARMONIE-AROME and the product can lead to spurious biases.

Positive effects of applying a sea ice data assimilation procedure in HARMONIE-AROME can be also found in the near-surface atmospheric variables, for example, screen-level temperature. Assessing the screen-level temperature RMSE computed for the first 12 hours of model forecasts initialised at OO UTC using observations from Svalbard and Franz Josef Land stations shows a considerable reduction of the error in the sensitivity experiment III^{AC}, compared to the reference model configuration III REF2 which does not constrain the ice state with observations (see the Figure 3.6). This reduction is more apparent for the stations located close to, or surrounded by, ice-covered areas where it reaches up to 5% on average. For the Svalbard stations located further away from the ice edge effects of applying sea ice analysis are less pronounced and for three stations



Figure 3.6: Relative change in the RMSE of the screen-level air temperature forecast resulting from applied sea ice data assimilation procedure in HARMONIE-AROME. RMSE is computed for up to 12 hour forecast lead time against in-situ observations.

tions the screen-level temperature RMSE of the sensitivity experiment is increased by 1% compared to the reference run. Similarly, for the station located in the marginal ice zone, applying sea ice data assimilation procedure leads to a considerable 5% increase of the air temperature RMSE. This reduction of the model forecast skill is attributed to reduced quality of the ice concentration field of HARMONIE-AROME near the ice edge combined with a similar feature of the assimilated ice surface temperature product.

Verifying the model forecasts of the nearsurface variables as a function of the forecast lead time by combining all the SYNOP stations within the model domain shows that sea ice data assimilation helps to improve the screen-level temperature forecast for the forecast lead time of up to 24 hours. At the o hour forecast lead time the experiments **III**^{AC}_{REF2} and $\mathbf{HI}_{\beta}^{\mathbf{AC}}$ show screen-level air temperature RMSE of 4.07 and 3.8°C, respectively. In both experiments RMSE grows as a function of the forecast lead time, however in the sensitivity experiment **III**^{AC} RMSE grows faster and difference between the two experiments diminishes. After 24 hours of model forecast the screen-level temperature RMSE reaches 4.28 and 4.26°C in \mathbf{HI}_{REF2}^{AC} and \mathbf{HI}_{β}^{AC} , respectively. Similar analysis performed for screen-level specific humidity and 10 m wind speed shows no clear impact of constraining the sea ice state in the model (see the Figure 9 in Paper III).

Improving the ice surface temperature in the model by applying a data assimilation procedure has a direct effect on the near-surface variables over sea ice and in the adjacent regions. However, improved ice surface temperature can also induce indirect positive effects on the upper atmosphere in the model through the atmospheric data assimilation if the observation operator of the atmospheric data assimilation system uses surface temperature field as one of its input parameters. Thus, an additional experiment **III**^{AC}_γ was conducted with providing the observation operator of the atmospheric data assimilation system of HAR-MONIE-AROME⁶⁶ with updated surface temperature field available after performing the surface analysis. A preliminary assessment of this effect showed that applying sea ice data assimilation procedure does not result in a clear positive impact on the model forecast of the upper atmosphere. However, updating the atmospheric data assimilation system to use the surface temperature analysis field as the model background in the observation opera-

tor

tor results in a response from the atmospheric model thus indicating potential benefits of this approach, which resembles a strongly coupled data assimilation procedure. For example, the difference between the reference experiment \mathbf{III}_{REF2}^{AC} and the experiment $\mathbf{III}_{\gamma}^{AC}$, which uses the updated surface temperature field in the observation operator of HARMONIE-AROME's 3DVAR data assimilation procedure, is greater than the difference between III_{REF2} and the experiment III^{AC}. Additionally, providing the updated surface temperature field to the observation operator, increases period of significant RMSE reduction of the screen-level temperature forecast in $\mathbf{III}_{\gamma}^{AC}$ to 12 hours compared to 5 hours in **III**^{AC}, which also indicates potential benefits of this solution.

3.4 Atmospheric response to small-scale features in the sea-ice cover

Numerical experiments and operational applications of HARMONIE-AROME use model grids with typical horizontal resolution of 2.5 km. This relatively high resolution allows explicitly representing some of the small-scale features of the ice cover, such as individual leads⁶⁷ and ice floes;^{68,69} for example, in the marginal ice zone. However, ice concentration products retrieved from passive microwave satellite data and applied to define the ice covered areas in HARMONIE-AROME usually have nominal spatial resolution of $\mathcal{O}(10 \text{ km})^{70,71}$ and even lower effective resolution.

Potential effects of applying high-resolution sea ice cover in regional NWP were assessed through a conservative modification of the ice concentration fields within the marginal ice zone in HARMONIE-AROME. It was found that presence of small-scale features in the experiment $\mathbf{IV}_{\alpha}^{AA}$ can have a well-pronounced effect on the near-surface atmospheric variables. Modifying the ice cover of HAR-MONIE-AROME to mimic the highly irregular structure of the real marginal ice zone results in a strong turbulent heat flux originated from introducing openings in the ice cover. However, the warm air plumes rising from these openings in the ice cover can not be fully reproduced by HARMONIE-AROME due to the coarse resolution of the model grid and simplified model physics. The turbulent flux can reach the values of up to 200 Wm⁻² higher in the experiment $\mathbf{IV}_{\alpha}^{AA}$ than in the experiment IV_{REF}^{AA} with unmodified ice cover. This heat flux induces a response in the atmospheric model, which can be traced in the standard deviation of the difference in surface heat flux computed when comparing the reference experiment IV_{REF}^{AA} and the sensitivity experiment $\mathbf{IV}_{\alpha}^{AA}$. The computed standard deviation field, largest over the modified marginal ice zone with the values reaching 100 Wm⁻² on average, also has high values over open sea, indicating that signals induced by the modified surface heat flux are advected to the regions located at the considerable distance downstream from the areas with modified ice cover.

Figure 3.7 shows similar features observed in the standard deviation of screen-level temperature and 10 m wind speed difference between the experiments \mathbf{IV}_{REF}^{AA} and $\mathbf{IV}_{\alpha}^{AA}$. Both these fields indicate that signals induced by the modified marginal ice zone in the experiment $\mathbf{IV}_{\alpha}^{\mathbf{AA}}$ propagate as far as the northern Norway, which is located about 1000 km away from the ice edge. However, response in the screenlevel air temperature field tends to be considerably weaker over large open water regions of the model domain, compared to the response in the 10 m wind speed. This behaviour is attributed to the peculiar implementation of the open sea parameterisation scheme⁷² of sur-FEX



Figure 3.7: Non-local effects in the screen-level air temperature and 10 m wind speed forecast of HARMONIE-AROME presented as a standard deviation of difference between six-hourly output of $\mathbf{IV}_{\alpha}^{AA}$ and \mathbf{IV}_{REF}^{AA} over the 3.5 month experiment. Also on the figure, location of the marginal ice zone, where kilometre-scale ice features were introduced in $\mathbf{IV}_{\alpha}^{AA}$ during the model experiments, is highlighted.

FEX used in HARMONIE-AROME. This parameterisation scheme does not include a prognostic model of the sea surface temperature and only parameterises the turbulent exchange between the ocean surface and the atmosphere while the sea surface temperature field is prescribed by means of an external data set, which is provided by the host model in case of the discussed experiments. Thus, when air masses are advected over the open water parts of the model domain, the prescribed sea surface temperature field tends to behave like a filter and diminishes the anomalies in the lowest model levels. Conversely, for the 10 m wind speed atmospheric response over open sea was considerably more pronounced, with no strong filtering effect from the ocean surface.

These findings indicate the potential benefits of applying high-resolution sea ice cover in operational regional NWP systems. Additionally, the response in the atmospheric system to the modifications of the ice cover field suggests it as a candidate for applying perturbations in operational ensemble prediction systems.

Chapter 4 Discussion

Sea ice cover in an NWP system operating in the Arctic can be represented with different level of detail, from a simple prescribed temperature field to a complex model with a dedicated data assimilation system. As shown in the Chapter 3, the model atmosphere of HAR-MONIE-AROME is sensitive to changes in the representation of the lower boundary condition over the ice-covered areas, which results in a pronounced effect on the forecast quality.

Applying even a simplified parameterisation of the sea ice cover in a high-resolution limited area NWP system, compared to prescribing the ice surface temperature, can be beneficial for operational applications. The study performed in the **Paper I** shows that introducing a simple thermodynamic sea ice scheme improves the representation of the ice surface temperature in the system. However, extended comparisons of performance of the sea ice scheme in the operational application of HARMONIE-AROME suggests that for very short-range forecasts a high-quality prescribed ice surface temperature field in HAR-MONIE-AROME can potentially outperform a simplified sea ice parameterisation scheme. Such behaviour of the model system is attributed to applying an over-simplified sea ice parameterisation scheme which represented a

snow-free ice layer of prescribed and uniform thickness. Conversely, for the forecasts longer than 12 hours using a parameterisation scheme, which can provide prognostic ice surface temperature, has clear advantages over prescribing the surface temperature field in the model. Verification against the land-based SYNOP stations shows that applying a prognostic sea ice scheme leads to a positive impact on the forecast skill of the near-surface variables, primarily the screen-level temperature, although the NWP system becomes sensitive to the quality of the prescribed sea ice cover, which tends to be lower in the areas with complex coast line such as Svalbard fjords. From the other side, the 10 m wind speed forecast in HAR-MONIE-AROME shows an increased ME when binary ice cover is replaced by the fractional ice cover, which can be compensated by taking into account the form drag when computing the drag coefficients over an ice-covered grid cell.

Performance of HARMONIE-AROME with a sea ice parameterisation scheme strongly depends on the level of simplification in the scheme. For example, **Paper I** uses a scheme which represents a snow-free surface with prescribed ice thickness of 0.75 m which results in a considerable positive bias of the ice surface

face temperature in the model. Applying a prescribed climatological ice thickness in the model reduces this bias but not completely. Initial experiments with representing the snow cover on top of the ice conducted in Paper I indicated potential benefits of extending the sea ice scheme by incorporating a snow model, although the results suggest insufficient spinup time in the performed simulations. A more thorough study in Paper II shows that missing snow cover, which is a common feature of many contemporary reanalysis products, results in misrepresented surface energy budget in the system causing a noticeable warm bias of the ice surface temperature, especially during clear-sky events. Increasing the prescribed ice thickness in a snow-free parameterisation scheme can not completely compensate the missing snow layer, particularly in case of binary ice cover in the model. Thus, a regional NWP system operating over highlatitude model domains can strongly benefit from applying a thermodynamic sea ice scheme with explicit representation of the ice thickness and snow cover evolution. However, such sea ice scheme requires a considerable spin-up period (preferably with the model cold start date being close to the September minimum of the sea ice extent in the Arctic) if initialisation data sets for snow cover over sea ice are not available.

Utilising a more detailed sea ice model helps to improve the representation of the ice cover in the NWP system, however a onedimensional scheme can not represent all the processes governing the evolution of the ice cover since it misses the ice dynamics. This deficiency can be compensated to some extent by constraining the sea ice state in the scheme by an observational product. **Paper III** assesses this approach by assimilating a L2 near real

time ice surface temperature product in HAR-MONIE-AROME. Constraining the ice surface temperature in the model helps to reduce the RMSE of the ice surface temperature in HAR-MONIE-AROME after analysis, although this improvement is short-lived and quickly diminishes as a function of the forecast lead time. For the SYNOP stations results are also positive in general with improvement of the screen-level temperature forecast in terms of RMSE lasting for up to 24 hours, although some stations, for example, located near the ice edge, show degraded forecast scores suggesting that this region can be problematic for the applied analysis scheme. Additionally, assimilating a nongap-free product results in strongly uneven observation coverage. Considering relatively high uncertainty of the ice surface temperature computed by a simplified one-dimensional parameterisation scheme further studies on the potential benefits of assimilation daily aggregated products should be conducted.

Timeliness of the near real time satellite products is a limiting factor for utilising them in operational applications. Introducing the so-called long cut-off cycle is a practical means to alleviating these constraints in an operational NWP system by the cost of performing an additional analysis and forecast each cycle. Effects of applying the presented sea ice data assimilation procedure in a long cut-off cycle under the timeliness constraints of operational applications of HARMONIE-AROME were not discussed in the present work and require further investigations.

Applying a sea ice data assimilation procedure opens a possibility of providing the updated ice surface temperature field to the atmospheric data assimilation procedure. Potentially, this approach can partially represent some effects of a strongly-coupled data assimilation ilation system without costly modifications within both surface and atmospheric data assimilation components of an NWP system. Initial study performed within the Paper III shows that HARMONIE-AROME gives a noticeable response in the model forecast to updating the surface temperature field in the model background of the atmospheric data assimilation procedure. The identified response was not assessed for giving positive or negative impact on the quality of the model forecast due to lack of the observational data and small size of the experimental model domain. A more detailed study utilising a larger model domain is required to investigate the feasibility of applying the suggested approach in the operational environment.

Relatively high resolution of contemporary regional NWP systems allows for more and more detail of the ice cover to be explicitly represented. With increasing spatial resolution sea ice cover becomes less and less smooth and individual leads and openings in the ice result in the strong local turbulent heat fluxes from the ocean to the atmosphere. **Paper IV** investigated the consequences of applying a high-resolution sea ice cover in a regional NWP system in an idealised environment. The study found that modifying the marginal ice zone in a conservative manner, while preserving the total ice area within the model domain, results in both local and non-local responses from the atmospheric system. The non-local response in the screen-level air temperature and 10 m wind speed reaching all the way from the ice edge to northern Norway suggests that accurate representation of the high-resolution marginal ice zone in a high-resolution NWP system can potentially benefit the quality of the weather forecast. However, since misplaced ice leads could result in misrepresented turbulent exchange between surface and the atmosphere, and position of individual leads and ice floes is very dynamic, benefits of prescribing the lead positions in a deterministic NWP system are not clear. This issue is expected to become more pressing as developments pursue higher and higher spatial resolution of regional NWP systems?³⁻⁷⁵ A more viable approach could be using the sea ice leads products for developing sea ice perturbations in an ensemble prediction system. In this case, misplacement errors in high-resolution sea-ice cover of each ensemble member would become less important.

Chapter 5 Conclusions

Sea ice in the modern-day Arctic still covers a great part of the Arctic Ocean. Presence of the ice cover insulates the ocean surface from the atmosphere while high albedo of the snowcovered ice reduces the amount of absorbed shortwave radiation. As a result, the state of the sea ice cover is an important factor governing the weather and climate in the Arctic. Recent drastic changes in the Arctic climate and diminishing of the ice cover result in increased economic activity in the region and this activity is expected to grow in the coming years?^{6,77} With increasing economic activity a higher demand is placed on the accurate and reliable weather forecasts?⁸ However, in the contemporary short-range regional NWP systems sea ice is often represented by simplified one-dimensional parameterisation schemes or even by prescribing the ice surface temperature field²⁴ This thesis is devoted to implementing a one-dimensional sea ice parameterisation scheme in the high-resolution regional NWP system HARMONIE-AROME²⁹ and assessing the effects induced by applying a simplified scheme on the quality of model forecasts in the Arctic.

The present work introduces a new onedimensional sea ice parameterisation scheme SICE within the surface parameterisation pack-

age of HARMONIE-AROME. SICE is configurable and allows representing the sea ice cover with various level of detail: from a snow-free ice surface with prescribed and uniform ice thickness, to explicitly modelled evolution of the snow cover and ice mass balance. Further, the thermal profile of sea ice in the scheme can be constrained by means of assimilating a remote-sensing observational product. A series of numerical experiments with applying different configurations of SICE in HAR-MONIE-AROME was conducted to improve the understanding of the effects induced by the sea ice scheme in the atmospheric forecast. Additionally, sea ice parameterisation schemes of several contemporary atmospheric reanalyses were studied. These schemes have different levels of complexity, allowing to assess the effects of applying a simplified model on the quality of the ice surface temperature in further detail.

The findings presented in Chapter 3 show that applying a prognostic sea ice parameterisation scheme has a positive effect on the model forecast of near-surface variables resulting from the reduced error growth rate of the ice surface temperature in the system. This effect becomes more pronounced with increasing forecast length, whilst in operational NWP systems with short forecast lengths, for example ple nowcasting systems, a high-quality prescribed ice surface temperature can outperform a simplified prognostic sea ice parameterisation scheme.

When a sea ice scheme does not explicitly represent the snow layer on top of the ice, the ice surface temperature in the model tends to have a considerable positive bias. Misrepresenting the ice thickness, for example by replacing it with a prescribed and uniform field, also leads to biases in the ice surface temperature. Although, in the presence of the snow cover contribution of errors in the ice thickness to the evolution of the ice surface temperature in the model is greatly diminished. When sea ice scheme utilises fractional sea ice cover, the quality of the sea ice concentration data becomes one of the most important parameters affecting the energy exchange between icecovered grid cells and the model atmosphere. For example, missing the land-fast ice over the Svalbard fjords results in a considerable warm bias in the screen-level air temperature forecast. From the other side, applying binary ice cover can result in a considerable negative bias of the ice surface temperature in the regions with low ice compactness.

Applying a data assimilation procedure to constrain the sea ice state in the system can help to further reduce the errors in the ice surface temperature, however the observed positive effect is rather short-lived and already greatly reduced after three hours of model forecast. Nevertheless, the impact on the screen-level air temperature forecast is pronounced for extended forecast lengths.

Resolution of the contemporary regional NWP systems allows for explicit representation of the small-scale features in the sea ice cover. HARMONIE-AROME shows a pronounced non-local response to presence of small scale features in the sea ice cover. Observed effects suggest that perturbing sea ice cover in the highly uncertain marginal ice zone can be beneficial for ensemble prediction systems operating in the Arctic.

This thesis presented a simple onedimensional sea ice parameterisation scheme suitable for operational short-range regional NWP applications. However, the main limitation of such schemes, absence of the ice dynamics, reduces potential benefits of applying them in NWP systems. From the other side, coupling with a complex dynamic-thermodynamic sea ice model could introduce unwanted complexity and require considerable development efforts. Some alternative approach, for example, modifying prognostic fields of a one-dimensional sea ice parameterisation scheme according to an externally provided sea ice drift data set, could be a pragmatic short-term solution, however, feasibility of such model configurations requires further investigation.

Problems of coupling a short-range regional NWP system with complex dynamicthermodynamic sea ice models, as well as potential benefits of such coupled systems over applying simplified one-dimensional parameterisation schemes, were not discussed in the present work. The next logical step would be assessing the applicability limits of simplified parameterisation schemes and their performance compared to dynamic-thermodynamic sea ice models in the context of short-range operational NWP. However, implementing and tuning a high-resolution regional coupled seaice-atmosphere NWP system is out of scope of the present work and these studies are left to the future research.

Chapter 6 Summary of the original publications

I. Batrak, Y., Kourzeneva, E. & Homleid, M. Implementation of a simple thermodynamic sea ice scheme, SICE version 1.0-38h1, within the ALADIN–HIRLAM numerical weather prediction system version 38h1. *Geoscientific Model Development* **11**, 3347–3368. DOI: 10.5194/gmd–11–3347–2018 (2018)

Paper I discusses the problem of parameterisation of sea ice in a regional high-resolution NWP system. A new sea ice parameterisation scheme is implemented and validated against in-situ and remote-sensing observational data sets. Numerical experiments show that applying the newly developed sea ice scheme, SICE, reduces the mean absolute error of the screenlevel air temperature forecast by 0.5 °C, but increases the wind speed bias by 0.4 ms⁻¹ on average in the areas close to sea ice. Validating the performance of HARMONIE-AROME with SICE against the satellite ice surface temperature products shows reduced error growth rates. However, the snow-free configuration of the scheme tends to have a considerable warm bias of the ice surface temperature.

The author developed and implemented SICE in HARMONIE-AROME, ran the major part of the numerical experiments, performed the verification of the model forecasts against the satellite ice surface temperature products, and wrote a major part of the manuscript.

II. Batrak, Y. & Müller, M. On the warm bias in atmospheric reanalyses induced by the missing snow over Arctic sea-ice. *Nature Communications* **10**, 4170. ISSN: 2041–1723. DOI: 10.1038/s41467-019-11975-3 (Sept. 2019)

Paper II investigates the performance of various contemporary atmospheric reanalysis products in terms of quality of the represented ice surface temperature. Validating the reanalysis products and two experimental configurations of HARMONIE-AROME against the data from a drift campaign shows that during Arctic winter clear-sky events missing representation of the snow layer in sea ice parameterisation schemes leads to a considerable 5–10°C warm bias of the modelled ice surface temperature. Additionally, detrimental effects of using

а

43

a binary representation of the sea ice cover in an NWP system, as well as relative contributions to the errors in the ice surface temperature of the misrepresented ice thickness and snow depth over sea ice are discussed. prognostic formulations for the evolution of the ice thickness within the SICE scheme, ran the numerical experiments, contributed to the analysis of the results and writing of a considerable part of the manuscript.

The author developed and implemented

III. Batrak, Y. Implementation of an Adaptive Bias-Aware Extended Kalman Filter for Sea-Ice Data Assimilation in the HARMONIE-AROME Numerical Weather Prediction System. *Journal of Advances in Modeling Earth Systems* **13**, e2021MS002533. DOI: 10.1029/ 2021MS002533 (2021)

Paper III implements a new data assimilation procedure in the surface parameterisation package of HARMONIE-AROME for sea ice data assimilation. Initial assessment shows that constraining the sea ice model by assimilating a satellite observational product helps to reduce the errors in the modelled ice surface temperature. The reduction in ice temperature RMSE is strongest right after the analysis, when it reaches 0.4 °C on average, and quickly diminishes as a function of the forecast lead time. However, these effects are still traceable after 3 hours of model forecast. Positive effect of sea ice data assimilation is also found in the screen-level air temperature forecast over the Svalbard and Franz Josef Land archipelagos. A new approach of using the output from surface analysis in atmospheric data assimilation of HARMONIE-AROME is discussed.

IV. Batrak, Y. & Müller, M. Atmospheric Response to Kilometer-Scale Changes in Sea Ice Concentration Within the Marginal Ice Zone. *Geophysical Research Letters* **45**, 6702–6709. DOI: 10.1029/2018GL078295 (2018)

Paper IV studies the response of an atmospheric NWP system to presence of fine-scale features in the sea ice cover. Numerical simulations show that applying conservative modifications within the marginal ice zone by introducing artificial openings in the ice cover results in both local and non-local responses in the atmospheric model. A pronounced atmospheric response with the 2 °C and 2 ms⁻¹ standard deviation of the differences in air temperature and wind speed, respectively, reaching as far as 500–1000 km from the ice edge indicates a potential benefit of applying highresolution sea ice cover in operational NWP applications, ensemble and deterministic.

The author performed the numerical experiments discussed in the study, contributed to the analysis of the results and writing of a considerable part of the manuscript.

Bibliography

- Kouraev, A., Papa, F., Mognard, N. M., Buharizin, P., Cazenave, A., Cretaux, J.-F., Dozortseva, J. & Remy, F. Sea ice cover in the Caspian and Aral Seas from historical and satellite data. *Journal of Marine Systems* 47. The Dying Aral Sea, Selected Papers from the 35th International Liege Colloquium on Ocean Dynamics, 89–100. DOI: 10.1016/ j.jmarsys.2003.12.011 (2004).
- 2. Yan, Y., Uotila, P., Huang, K. & Gu, W. Variability of sea ice area in the Bohai Sea from 1958 to 2015. *Science of The Total Environment* **709**, 136–164. DOI: 10.1016/j.scitotenv.2019.136164 (2020).
- 3. Arrigo, K. R. Sea Ice Ecosystems. *Annual Review of Marine Science* **6**, 439–467. DOI: 10.1146/annurev-marine-010213-135103 (2014).
- 4. Budikova, D. Role of Arctic sea ice in global atmospheric circulation: A review. *Global and Planetary Change* 68, 149–163. DOI: 10.1016/j.gloplacha.2009.04.001 (2009).
- Gao, Y., Sun, J., Li, F., He, S., Sandven, S., Yan, Q., Zhang, Z., Lohmann, K., Keenlyside, N., et al. Arctic sea ice and Eurasian climate: A review. *Advances in Atmospheric Sciences* 32, 92–114. DOI: 10.1007/s00376-014-0009-6 (Jan. 2015).
- 6. Vihma, T. Effects of Arctic Sea Ice Decline on Weather and Climate: A Review. *Surveys in Geophysics* **35**, 1175–1214. DOI: 10.1007/s10712-014-9284-0 (Sept. 2014).
- Knies, J., Cabedo-Sanz, P., Belt, S. T., Baranwal, S., Fietz, S. & Rosell-Melé, A. The emergence of modern sea ice cover in the Arctic Ocean. *Nature Communications* 5, 5608. DOI: 10.1038/ncomms6608 (Nov. 2014).
- Mansfield, D. A. Polar lows: The development of baroclinic disturbances in cold air outbreaks. *Quarterly Journal of the Royal Meteorological Society* 100, 541–554. DOI: 10.1002/qj.49710042604 (1974).
- 9. Kolstad, E. W., Bracegirdle, T. J. & Seierstad, I. A. Marine cold-air outbreaks in the North Atlantic: temporal distribution and associations with large-scale atmospheric circulation.

Climate Dynamics **33**, 187–197. ISSN: 1432-0894. DOI: 10.1007/s00382-008-0431-5 (Aug. 2009).

- Nordeng, T. E., Brunet, G. & Caughey, J. Improvement of Weather Forecasts in Polar Regions. WMO Bulletin 56, 250–257 (2007).
- Jung, T., Gordon, N. D., Bauer, P., Bromwich, D. H., Chevallier, M., Day, J. J., Dawson, J., Doblas-Reyes, F., Fairall, C., *et al.* Advancing Polar Prediction Capabilities on Daily to Seasonal Time Scales. *Bulletin of the American Meteorological Society* 97, 1631–1647. DOI: 10.1175/BAMS-D-14-00246.1 (2016).
- 12. ACIA. Arctic Climate Impact Assessment. ACIA Overview report (Cambridge University Press, Cambridge, May 2005).
- Richter-Menge, J., Overland, J., Proshutinsky, A., Romanovsky, V., Bengtsson, L., Brigham, L., Dyurgerov, M., Gascard, J., Gerland, S., *et al. State of the Arctic Report.* NOAA OAR Special Report tech. rep. (NOAA/OAR/PMEL, Seattle, WA, Oct. 2006).
- 14. Thoman, R. L., Richter-Menge, J. & Druckenmiller, M. L. NOAA Arctic Report Card 2020 Executive Summary Administrative Report. DOI: 10.25923/mn5p-t549 (2020).
- 15. Gilbert, H. *A discourse of a discoverie for a new passage to Cataia* (Henry Middleton for Richarde Jhones, London, Kingdom of England, 1576).
- 16. Lomonosov, M. A short description of various North Seas expeditions and an evincement of a possible passage to the East Indies by the Siberian Ocean [Краткое описание разныхъ путешествій по съвернымъ морямъ и показание возможнаго проходу Сибирскимъ Океаномъ въ восточную Индію] Presented to His Imperial Highness the Successor Tsesarevich and Grand Duke Paul Petrovich (Saint Petersburg, Russian Empire, 1763).
- Brown, J. The North-West Passage, and the Plans for the Search for Sir John Franklin: A Review with Maps 2nd ed. (E. Stanford, London, United Kingdom of Great Britain and Ireland, 1860).
- Kogan, E. The voyage of the expedition ship Hertha in search of lieutenant Brusilov and his companions in 1915 (preliminary report) [Плаваніе экспедиціоннаго судна "Герта" для поисковъ лейтенанта Брусилова и его спутниковъ въ 1915 г. (Предварительный отчетъ)] 42 (printing house of the Naval Ministry, Petrograd, Russian Empire, 1916).
- Barr, W. The fate of Rusanov's Gerkules expedition in the Kara Sea, 1913; Some Further details and recent developments. *Polar Record* 22, 287–304. DOI: 10.1017/S0032247400005416 (1984).
- 20. Crépin, A.-S., Karcher, M. & Gascard, J.-C. Arctic Climate Change, Economy and Society (ACCESS): Integrated perspectives. *Ambio* **46**, 341–354. ISSN: 1654-7209. DOI: 10 . 1007/s13280-017-0953-3 (Dec. 2017).

- WMO. Guide to Instruments and Methods of Observation 2018 edition, 84–85. ISBN: 978-92-63-10008-5 (World Meteorological Organization (WMO), Geneva, Switzerland, 2018).
- 22. Weyprecht, K. *Die Metamorphosen des Polareises* (Moritz Perles, Vienna, Austro-Hungarian Empire, 1879).
- 23. Stefan, J. Über die Theorie der Eisbildung, insbesondere über die Eisbildung im Polarmeere. Annalen der Physik 278, 269–286. DOI: 10.1002/andp.18912780206 (1891).
- Køltzow, M., Casati, B., Bazile, E., Haiden, T. & Valkonen, T. An NWP Model Intercomparison of Surface Weather Parameters in the European Arctic during the Year of Polar Prediction Special Observing Period Northern Hemisphere 1. *Weather and Forecasting* 34, 959–983. DOI: 10.1175/WAF-D-19-0003.1 (2019).
- Browne, P. A., de Rosnay, P., Zuo, H., Bennett, A. & Dawson, A. Weakly Coupled Ocean-Atmosphere Data Assimilation in the ECMWF NWP System. *Remote Sensing* 11. ISSN: 2072–4292. DOI: 10.3390/rs11030234 (2019).
- Guiavarc'h, C., Roberts-Jones, J., Harris, C., Lea, D. J., Ryan, A. & Ascione, I. Assessment of ocean analysis and forecast from an atmosphere–ocean coupled data assimilation operational system. *Ocean Science* 15, 1307–1326. DOI: 10.5194/0s-15-1307-2019 (2019).
- Meixner, J., Stefanova, L., Worthen, D., Wang, J., Wang, J., Moorthi, S., Li, B., Kuang, J., Grumbine, R., *et al.* Model Development of the Unified Forecast System for Subseasonal to Seasonal Timescales. *Research Activities In Earth System Modelling. Report No. 50.* WCRP Report No. 6/2020, 9.03–9.04 (July 2020).
- 28. Schröder, D., Heinemann, G. & Willmes, S. The impact of a thermodynamic sea-ice module in the COSMO numerical weather prediction model on simulations for the Laptev Sea, Siberian Arctic. *Polar Research*. DOI: 10.3402/polar.v30i0.6334 (May 2011).
- 29. Bengtsson, L., Andrae, U., Aspelien, T., Batrak, Y., Calvo, J., de Rooy, W., Gleeson, E., Hansen-Sass, B., Homleid, M., *et al.* The HARMONIE–AROME Model Configuration in the ALADIN–HIRLAM NWP System. *Monthly Weather Review* **145**, 1919–1935. ISSN: 0027-0644. DOI: 10.1175/MWR-D-16-0417.1 (Apr. 2017).
- Masson, V., Le Moigne, P., Martin, E., Faroux, S., Alias, A., Alkama, R., Belamari, S., Barbu, A., Boone, A., *et al.* The SURFEXv7.2 land and ocean surface platform for coupled or offline simulation of earth surface variables and fluxes. *Geoscientific Model Development* 6, 929–960. DOI: 10.5194/gmd-6-929-2013 (2013).

- Boone, A. & Etchevers, P. An Intercomparison of Three Snow Schemes of Varying Complexity Coupled to the Same Land Surface Model: Local-Scale Evaluation at an Alpine Site. *Journal of Hydrometeorology* 2, 374–394. DOI: 10.1175/1525-7541(2001) 002<0374:AIOTSS>2.0.C0;2(2001).
- 32. Vihma, T., Pirazzini, R., Fer, I., Renfrew, I. A., Sedlar, J., Tjernström, M., Lüpkes, C., Nygård, T., Notz, D., *et al.* Advances in understanding and parameterization of smallscale physical processes in the marine Arctic climate system: a review. *Atmospheric Chemistry and Physics* **14**, 9403–9450. DOI: 10.5194/acp-14-9403-2014 (Sept. 2014).
- 33. Rösel, A., Itkin, P., King, J., Divine, D., Wang, C., Granskog, M. A., Krumpen, T. & Gerland, S. Thin Sea Ice, Thick Snow, and Widespread Negative Freeboard Observed During N-ICE2015 North of Svalbard. *Journal of Geophysical Research: Oceans* **123**, 1156–1176. DOI: 10.1002/2017 JC012865 (2018).
- Merkouriadi, I., Liston, G. E., Graham, R. M. & Granskog, M. A. Quantifying the Potential for Snow-Ice Formation in the Arctic Ocean. *Geophysical Research Letters* 47, e2019GL085020. DOI: 10.1029/2019GL085020 (2020).
- Lüpkes, C., Gryanik, V. M., Hartmann, J. & Andreas, E. L. A parametrization, based on sea ice morphology, of the neutral atmospheric drag coefficients for weather prediction and climate models. *Journal of Geophysical Research: Atmospheres* 117. DOI: 10.1029/ 2012JD017630 (2012).
- 36. Dybkjær, G., Eastwood, S. & Howe, E. *OSI SAF High Latitudes L2 Sea and Sea Ice Surface Temperature Product User Manual* tech. rep. (Danish Meteorological Institute and Norwegian Meteorological Institute, Mar. 2018).
- Dybkjær, G., Tonboe, R. & Høyer, J. L. Arctic surface temperatures from Metop AVHRR compared to in situ ocean and land data. *Ocean Science* 8, 959–970. DOI: 10.5194/0s– 8–959–2012 (2012).
- Vihma, T., Johansson, M. M. & Launiainen, J. Radiative and turbulent surface heat fluxes over sea ice in the western Weddell Sea in early summer. *Journal of Geophysical Research: Oceans* 114. DOI: 10.1029/2008JC004995 (2009).
- Earth science reference handbook: a guide to NASA's Earth science program and Earth observing satellite missions (eds Parkinson, C. L., Ward, A. & King, M. D.) 31–32 (Washington, D.C.: National Aeronautics and Space Administration, 2006).
- Hall, D. K. & Riggs, G. MODIS/Terra Sea Ice Extent 5-Min L2 Swath 1km, Version 6. Boulder, Colorado USA. NASA National Snow and Ice Data Center Distributed Active Archive Center 2015. DOI: 10.5067/MODIS/MOD29.006.

- Hall, D. K. & Riggs, G. MODIS/Aqua Sea Ice Extent 5-Min L2 Swath 1km, Version 6. Boulder, Colorado USA. NASA National Snow and Ice Data Center Distributed Active Archive Center 2015. DOI: 10.5067/MODIS/MYD29.006.
- Tschudi, M., Riggs, G., Hall, D. K. & Román, M. O. VIIRS/NPP Ice Surface Temperature 6-Min L2 Swath 750m, Version 1. NASA National Snow and Ice Data Center Distributed Active Archive Center 2017. DOI: 10.5067/VIIRS/VNP30.001.
- 43. CMEMS. Arctic Ocean Sea and Ice Surface Temperature (2021). DOI: 10.48670/moi-00130.
- 44. Granskog, M. A., Assmy, P., Gerland, S., Spreen, G., Steen, H. & Smedsrud, L. H. Arctic research on thin ice: Consequences of Arctic sea ice loss. *Eos* **97.** DOI: 10.1029/2016E0044097 (2016).
- Kahl, J. D., Serreze, M. C., Shiotani, S., Skony, S. M. & Schnell, R. C. In Situ Meteorological Society cal Sounding Archives for Arctic Studies. *Bulletin of the American Meteorological Society* 73, 1824–1830. DOI: 10.1175/1520-0477 (1992)073<1824: ISMSAF>2.0.C0;2 (1992).
- 46. Naakka, T., Nygård, T., Tjernström, M., Vihma, T., Pirazzini, R. & Brooks, I. M. The Impact of Radiosounding Observations on Numerical Weather Prediction Analyses in the Arctic. *Geophysical Research Letters* **46**, 8527–8535. DOI: 10.1029/2019GL083332 (2019).
- Gustafsson, N., Janjić, T., Schraff, C., Leuenberger, D., Weissmann, M., Reich, H., Brousseau, P., Montmerle, T., Wattrelot, E., *et al.* Survey of data assimilation methods for convective-scale numerical weather prediction at operational centres. *Quarterly Journal of the Royal Meteorological Society* 144, 1218–1256. DOI: 10.1002/qj.3179 (2018).
- Bengtsson, L. & Shukla, J. Integration of Space and In Situ Observations to Study Global Climate Change. *Bulletin of the American Meteorological Society* 69, 1130–1143. DOI: 10.1175/1520-0477(1988)069<1130:IOSAIS>2.0.C0;2(1988).
- 49. Parker, W. S. Reanalyses and Observations: What's the Difference? *Bulletin of the American Meteorological Society* 97, 1565–1572. DOI: 10.1175/BAMS-D-14-00226.1 (2016).
- Gelaro, R., McCarty, W., Suárez, M. J., Todling, R., Molod, A., Takacs, L., Randles, C. A., Darmenov, A., Bosilovich, M. G., *et al.* The Modern-Era Retrospective Analysis for Research and Applications, Version 2 (MERRA-2). *J. Climate* **30**, 5419–5454. DOI: 10.1175/JCLI-D-16-0758.1 (2017).
- 51. Dee, D. P., Uppala, S. M., Simmons, A. J., Berrisford, P., Poli, P., Kobayashi, S., Andrae, U., Balmaseda, M. A., Balsamo, G., *et al.* The ERA-Interim reanalysis: configuration and

performance of the data assimilation system. *Quarterly Journal of the Royal Meteorological Society* **137**, 553–597. ISSN: 1477-870X. DOI: 10.1002/qj.828 (2011).

- Hersbach, H., Bell, B., Berrisford, P., Hirahara, S., Horányi, A., Muñoz-Sabater, J., Nicolas, J., Peubey, C., Radu, R., et al. The ERA5 global reanalysis. *Quarterly Journal of the Royal Meteorological Society* 146, 1999–2049. DOI: 10.1002/qj.3803 (2020).
- 53. Kobayashi, S., Ota, Y., Harada, Y., Ebita, A., Moriya, M., Onoda, H., Onogi, K., Kamahori, H., Kobayashi, C., *et al.* The JRA-55 Reanalysis: General Specifications and Basic Characteristics. *J. Meteorol. Soc. Jpn. Ser. II* **93**, 5–48. DOI: 10.2151/jmsj.2015-001 (2015).
- 54. Kanamitsu, M., Ebisuzaki, W., Woollen, J., Yang, S.-K., Hnilo, J. J., Fiorino, M. & Potter, G. L. NCEP-DOE AMIP-II reanalysis (R-2). *Bull. Amer. Math. Soc.* 83, 1631–1644. DOI: 10.1175/BAMS-83-11-1631 (2002).
- Müller, M., Batrak, Y., Kristiansen, J., Køltzow, M. A. Ø., Noer, G. & Korosov, A. Characteristics of a Convective-Scale Weather Forecasting System for the European Arctic. *Monthly Weather Review* 145, 4771–4787. DOI: 10.1175/MWR-D-17-0194.1 (2017).
- Müller, M., Homleid, M., Ivarsson, K.-I., Køltzow, M. A. Ø., Lindskog, M., Midtbø, K. H., Andrae, U., Aspelien, T., Berggren, L., *et al.* AROME-MetCoOp: A Nordic Convective-Scale Operational Weather Prediction Model. *Weather an Forecasting* 32, 609–627. DOI: 10.1175/WAF-D-16-0099.1 (2017).
- Randriamampianina, R., Bormann, N., Køltzow, M. A. Ø., Lawrence, H., Sandu, I. & Wang, Z. Q. Relative impact of observations on a regional Arctic numerical weather prediction system. *Quarterly Journal of the Royal Meteorological Society* 147, 2212–2232. DOI: 10.1002/qj.4018 (2021).
- Taillefer, F. CANARI—technical documentation—based on ARPEGE cycle CY25T1 (AL25T1 for ALADIN) tech. rep. (Météo-France, 2002).
- 59. ECMWF. in *IFS Documentation CY40R1 IFS Documentation* 4. Operational implementation 22 November 2013 (ECMWF, 2014). DOI: 10.21957/f56vvey1x.
- 60. Rinke, A., Maslowski, W., Dethloff, K. & Clement, J. Influence of sea ice on the atmosphere: A study with an Arctic atmospheric regional climate model. *Journal of Geophysical Research: Atmospheres* **111.** DOI: 10.1029/2005JD006957 (2006).
- Hines, K. M., Bromwich, D. H., Bai, L., Bitz, C. M., Powers, J. G. & Manning, K. W. Sea Ice Enhancements to Polar WRF. *Monthly Weather Review* 143, 2363–2385. DOI: 10.1175/MWR-D-14-00344.1 (2015).
- 62. Maykut, G. A. & Untersteiner, N. Some results from a time-dependent thermodynamic model of sea ice. *Journal of Geophysical Research (1896-1977)* **76,** 1550–1575. DOI: 10. 1029/JC076i006p01550 (1971).

- 63. Maykut, G. A. Energy exchange over young sea ice in the central Arctic. *Journal of Geophysical Research: Oceans* 83, 3646–3658. DOI: 10.1029/JC083iC07p03646 (1978).
- Graham, R. M., Cohen, L., Ritzhaupt, N., Segger, B., Graversen, R. G., Rinke, A., Walden, V. P., Granskog, M. A. & Hudson, S. R. Evaluation of Six Atmospheric Reanalyses over Arctic Sea Ice from Winter to Early Summer. *Journal of Climate* 32, 4121–4143. DOI: 10.1175/JCLI-D-18-0643.1 (2019).
- 65. Sorteberg, A., Kattsov, V., Walsh, J. E. & Pavlova, T. The Arctic surface energy budget as simulated with the IPCC AR4 AOGCMs. *Climate Dynamics* **29**, 131–156. ISSN: 1432-0894. DOI: 10.1007/s00382-006-0222-9 (Aug. 2007).
- Saunders, R., Hocking, J., Turner, E., Rayer, P., Rundle, D., Brunel, P., Vidot, J., Roquet, P., Matricardi, M., *et al.* An update on the RTTOV fast radiative transfer model (currently at version 12). *Geoscientific Model Development* 11, 2717–2737. DOI: 10.5194/gmd-11-2717-2018 (2018).
- 67. Lindsay, R. W. & Rothrock, D. A. Arctic sea ice leads from advanced very high resolution radiometer images. *Journal of Geophysical Research: Oceans* **100**, 4533–4544. DOI: 10. 1029/94JC02393 (1995).
- 68. Hudson, R. D. Multiyear sea ice floe distribution in the Canadian Arctic Ocean. *Journal* of *Geophysical Research: Oceans* **92**, 14663–14669. DOI: 10.1029/JC092iC13p14663 (1987).
- 69. Perovich, D. K. & Jones, K. F. The seasonal evolution of sea ice floe size distribution. *Journal of Geophysical Research: Oceans* **119**, 8767–8777. DOI: 10.1002/2014JC010136 (2014).
- Ivanova, N., Johannessen, O. M., Pedersen, L. T. & Tonboe, R. T. Retrieval of Arctic Sea Ice Parameters by Satellite Passive Microwave Sensors: A Comparison of Eleven Sea Ice Concentration Algorithms. *IEEE Transactions on Geoscience and Remote Sensing* 52, 7233–7246. DOI: 10.1109/TGRS.2014.2310136 (2014).
- Kern, S., Lavergne, T., Notz, D., Pedersen, L. T., Tonboe, R. T., Saldo, R. & Sørensen, A. M. Satellite passive microwave sea-ice concentration data set intercomparison: closed ice and ship-based observations. *The CryoSphere* 13, 3261–3307. DOI: 10.5194/tc-13-3261-2019 (2019).
- 72. Belamari, S. *Report on uncertainty estimates of an optimal bulk formulation for turbulent fluxes* tech. rep. D.4.1.2 (MERSEA IP Deliverable, 2005).
- 73. Køltzow, M., Grote, R. & Singleton, A. On the configuration of a regional Arctic Numerical Weather Prediction system to maximize predictive capacity. *Tellus A: Dynamic*

Meteorology and Oceanography **73**, 1–18. DOI: 10.1080/16000870.2021.1976093 (2021).

- 74. Valkonen, T., Stoll, P., Batrak, Y., Køltzow, M., Schneider, T. M., Stigter, E. E., Aashamar, O. B., Støylen, E. & Jonassen, M. O. Evaluation of a sub-kilometre NWP system in an Arctic fjord-valley system in winter. *Tellus A: Dynamic Meteorology and Oceanography* **72**, 1–21. DOI: 10.1080/16000870.2020.1838181 (2020).
- 75. Vionnet, V., Bélair, S., Girard, C. & Plante, A. Wintertime Subkilometer Numerical Forecasts of Near-Surface Variables in the Canadian Rocky Mountains. *Monthly Weather Review* **143**, 666–686. DOI: 10.1175/MWR-D-14-00128.1 (2015).
- 76. Meier, W. N., Hovelsrud, G. K., van Oort, B. E., Key, J. R., Kovacs, K. M., Michel, C., Haas, C., Granskog, M. A., Gerland, S., *et al.* Arctic sea ice in transformation: A review of recent observed changes and impacts on biology and human activity. *Reviews of Geophysics* 52, 185–217. DOI: 10.1002/2013RG000431 (2014).
- 77. Stocker, A. N., Renner, A. H. H. & Knol-Kauffman, M. Sea ice variability and maritime activity around Svalbard in the period 2012–2019. *Scientific Reports* 10, 17043. ISSN: 2045–2322. DOI: 10.1038/s41598-020-74064-2 (Oct. 2020).
- 78. Emmerson, C. & Lahn, G. Arctic opening: Opportunity and risk in the high north. *Chatham House-Lloyd's Risk Insight Report* (2012).